Author's response to reviews

Title: Variation in the CXCR1 gene (IL8RA) is not associated with susceptibility to chronic periodontitis

Authors:

Raquel M Scarel-Caminaga (raquel@foar.unesp.br)
Karen MC Curtis (karcurtis@gmail.com)
Rivelto Renzi (riveltorenzi@hotmail.com)
Patrícia Sogumo (paty_sogumo@hotmail.com)
Giovana Anovazzi (giovanaanovazzi@hotmail.com)
Aline C Viana (dentistaaline@yahoo.com.br)
Yeon J Kim (yeon_jungkim@yahoo.com.br)
Silvana RP Orrico (s-orrico@foar.unesp.br)
Joni A Cirelli (cirelli@foar.unesp.br)

Version: 3 Date: 1 November 2011

Author's response to reviews:

November, 1st, 2011

The requested major and minor revisions were made according to the Independent Editorial Production Team, as commented below:

Author list: The list of authors in the manuscript was rewritten exactly as the preferred style.

Abstract: The text was checked. The Greek symbol was replaced by the standard letter ‘alpha’. Also, the number of subjects was corrected.

Accession Numbers: The accession numbers of nucleic acid sequences, was provided in square brackets and include the corresponding database name, according to the example.

The tables were included in the manuscript in a tables section following the references. All visible vertical lines were removed from the tables.

Minor revisions

All highlighting and tracking were removed from the manuscript. All text is now black.

The typography was checked, as well as all the references.

Thank you

Raquel M Scarel-Caminaga